Author's response to reviews

Title: Giant hepatic hydatid cyst with sub-fascial extension treated by open minimally invasive surgery: a case report

Authors:

- Dipesh Duttaroy (drduttaroy@gmail.com)
- Samir Kacheriwala (dr_samir_k@yahoo.com)
- Bithika Duttaroy (drbithika@gmail.com)
- Jitendra Jagtap (drjitendrajagtap@gmail.com)
- Gunjan Patel (gunjanap@yahoo.com)
- Nikhil Modi (drnjmodi@yahoo.com)

Version: 5 Date: 16 January 2008

Author's response to reviews:

Dear Sir,

As per the formatting changes requested I have made the changes.

The author list, author email addresses and the abstract have been updated as per the journal requirements in the manuscript.

Thanking You

Dr D Duttaroy